Author’s response to reviews

Title: Vitamin E and telmisartan attenuates doxorubicin induced cardiac injury in rat through down regulation of inflammatory response

Authors:

Najah Hadi (drnajahhadi@yahoo.com)
nasser Ghaly Yousif (nasser.yousif@uni-ulm.de)
fadhil al-amran (fadhil.al-amran@ucdenver.edu)
Bassim Mohammad (jumabassim@yahoo.co.uk)
Sadiq J. Ali (drsadiq67@yahoo.co.uk)
Nadhem K. Huntei (najahriness@yahoo.com)

Version: 5 Date: 27 May 2012

Author's response to reviews:

Cover letter
Dear editors,
In addition to the previous changes we added the Competing interests, Author’s contribution and acknowledgement.

Nasser G. Yousif, M.D.
University of Colorado Denver
Anschutz Medical Campus RC2, Room 6410 12700 E 19th Ave
Aurora, CO 80045
Tel: (303)724-6318
Fax:(303)724-6330
Mobile:(586)246-6013